Author’s response to reviews

Title: Adjuvant chemotherapy versus chemoradiotherapy for small cell lung cancer with lymph node metastasis: a retrospective observational study with use of a national database in Japan

Authors:
Hirokazu Urushiyama (hirourushi@yahoo.co.jp)
Taisuke Jo (jo.taisuke2016@gmail.com)
Hideo Yasunaga (yasunagah-tky@umin.ac.jp)
Yasuhiro Yamauchi (yas-kkr@umin.ac.jp)
Hiroki Matsui (ptmatsui-tky@umin.ac.jp)
Wakae Hasegawa (wakaetnk0703@yahoo.co.jp)
Hideyuki Takeshima (Htakeshima-tky@umin.ac.jp)
Yoshihisa Hiraishi (blueeyedsole1980@gmail.com)
Akihisa Mitani (mitania-tky@umin.ac.jp)
Kiyohide Fushimi (kfushimi.hci@tmd.ac.jp)
Takahide Nagase (takahide-tky@umin.ac.jp)

Version: 2 Date: 21 Aug 2017

Author’s response to reviews:

We wish to thank the editors and reviewers for their careful reading of our manuscript and for their insightful comments, which we believe have helped us to improve our paper. We have revised the manuscript in accordance with the editor comments.